Author's response to reviews

Title: Changes in mental health services and suicide mortality in Norway: an ecological study

Authors:

Håkon A. Johannessen (hajn@fhi.no)
Gudrun Dieserud (gudrun.dieserud@fhi.no)
Bjørgulf Claussen (bjorgulf.claussen@medisin.uio.no)
Per-Henrik Zahl (per-henrik.zahl@fhi.no)

Version: 7 Date: 20 March 2011

Author's response to reviews:

Dear Dr Christopher B. Forrest,

I have made one typographical change on page 9, fourth paragraph. I have changed “adjusted incidence rate ratio” to “adjusted Incidence Rate Ratio”.

In addition, I would like you to check if the formatting of the tables is correct. The legends are both above and below the tables.

Best regards,

Mr. Håkon A. Johannessen